

G OPEN ACCESS

Citation: Mofiz E, Seemann T, Bahlo M, Holt D, Currie BJ, Fischer K, et al. (2016) Mitochondrial Genome Sequence of the Scabies Mite Provides Insight into the Genetic Diversity of Individual Scabies Infections. PLoS Negl Trop Dis 10(2): e0004384. doi:10.1371/journal.pntd.0004384

Editor: Joseph M. Vinetz, University of California San Diego School of Medicine, UNITED STATES

Received: September 6, 2015

Accepted: December 20, 2015

Published: February 12, 2016

Copyright: © 2016 Mofiz et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited..

Data Availability Statement: All data are available from the European Nucleotide Archive (accession numbers LN874268, LN874269, LN874270). Raw data is available via ENA accession PRJEB12428.

Funding: This research was supported by The Scobie and Claire Mackinnon Trust, The Lettisier Foundation and Evans Family Foundation, and the National Health and Medical Research Council (NHMRC) Program Grants (1054618 and 496600). ATP was supported by an NHMRC Career Development Fellowship (1003856). KF and MB were supported by Australian Research Council Future RESEARCH ARTICLE

Mitochondrial Genome Sequence of the Scabies Mite Provides Insight into the Genetic Diversity of Individual Scabies Infections

Ehtesham Mofiz^{1,2}, Torsten Seemann³, Melanie Bahlo^{1,4}, Deborah Holt⁵, Bart J. Currie⁵, Katja Fischer^{6©}, Anthony T. Papenfuss^{1,2,7,8©}*

 Bioinformatics Division, The Walter and Eliza Hall Institute of Medical Research, Parkville, Victoria, Australia, 2 Department of Medical Biology, University of Melbourne, Melbourne, Victoria, Australia,
Victorian Life Sciences Computation Initiative, University of Melbourne, Melbourne, Victoria, Australia,
Department of Mathematics and Statistics, University of Melbourne, Victoria, Australia, 5 Menzies School of Health Research, Charles Darwin University, Casuarina, Northern Territory, Australia, 6 QIMR Berghofer Medical Research, Herston, Queensland, Australia, 7 Sir Peter MacCallum Department of Oncology, University of Melbourne, Melbourne, Victoria, Australia, 8 Peter MacCallum Cancer Centre, East Melbourne, Australia

• These authors contributed equally to this work.

* papenfuss@wehi.edu.au

Abstract

The scabies mite, Sarcoptes scabiei, is an obligate parasite of the skin that infects humans and other animal species, causing scabies, a contagious disease characterized by extreme itching. Scabies infections are a major health problem, particularly in remote Indigenous communities in Australia, where co-infection of epidermal scabies lesions by Group A Streptococci or Staphylococcus aureus is thought to be responsible for the high rate of rheumatic heart disease and chronic kidney disease. We collected and separately sequenced mite DNA from several pools of thousands of whole mites from a porcine model of scabies (S. scabiei var. suis) and two human patients (S. scabiei var. hominis) living in different regions of northern Australia. Our sequencing samples the mite and its metagenome, including the mite gut flora and the wound micro-environment. Here, we describe the mitochondrial genome of the scabies mite. We developed a new de novo assembly pipeline based on a bait-and-reassemble strategy, which produced a 14 kilobase mitochondrial genome sequence assembly. We also annotated 35 genes and have compared these to other Acari mites. We identified single nucleotide polymorphisms (SNPs) and used these to infer the presence of six haplogroups in our samples, Remarkably, these fall into two closely-related clades with one clade including both human and pig varieties. This supports earlier findings that only limited genetic differences may separate some human and animal varieties, and raises the possibility of cross-host infections. Finally, we used these mitochondrial haplotypes to show that the genetic diversity of individual infections is typically small with 1–3 distinct haplotypes per infestation.

Fellowships. The research benefitted by support from the Victorian State Government Operational Infrastructure Support and Australian Government NHMRC Independent Research Institute Infrastructure Support. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Author Summary

The scabies mite is a skin parasite that infects humans and other animal species, causing scabies, a contagious disease characterized by extreme itching. Scabies infections are a major health problem in developing countries and in indigenous Australian populations, where scabies is associated with pyoderma (skin sores) and linked to rheumatic fever and rheumatic heart disease. Little is known about the genetics of the scabies mite. We have assembled the mitochondrial genome of scabies mites obtained from human patients in Australia and from a pig model. While investigating the genetic diversity of each infestation, we found that mitochrondial genomes clustered into two broad clades and showed limited genetic diversity within each infestation. Remarkably, one closely related clade included both human and pig mites, suggesting that mite transmission from pig to human may be possible. This could have major implications in the management of porcine mange and human scabies.

Introduction

The scabies mite is an ectoparasitic arachnid that causes an itchy skin infection, known as scabies. Each year around 300 million people worldwide are affected by scabies [1]. Scabies is responsible for a significant disease burden in affected populations through its obligate parasitic lifecycle, which facilitates secondary infections by other pathogens. A severe, but more rare form of scabies, known as crusted scabies, is characterised by hyper-infestation. It generally occurs in immune-compromised individuals [2], although it can occur in patients with no overt immunological deficiency [3]. Cases of crusted scabies can play a significant role in transmission [4]. The mite also infects more than a hundred species of mammals, creating an animal welfare and economic burden in primary industry [5–7].

Scabies represents a major health problem in many remote Indigenous communities in Australia and particularly affects children. Up to 25% of adults and 50% of children acquire scabies infections each year, and 7 out of 10 children under 1 year contract scabies with first presentation peaking at 2 months of age [4, 8]. Scabies is associated with pyoderma (skin sores) in Indigenous communities, but also in many other circumstances of disadvantage globally [9]. In tropical regions, the major pathogens of pyoderma are Group A streptococcus (Streptococcus pyogenes; GAS) and Staphylococcus aureus; with S. pyogenes considered the dominant and usually primary pathogen [4, 10, 11]. This is also the case for remote Indigenous communities in northern and central Australia [4, 10-12]. Sequelae of infection with GAS include acute poststreptococcal glomerulonephritis, which can be clustered (epidemic) or sporadic, and acute rheumatic fever [4, 13, 14]. Rheumatic heart disease, characterised by heart valve damage, occurs as a consequence of acute rheumatic fever and repeated episodes of it can result in cumulative heart valve damage, with consequent heart failure and death [5, 13]. The prevalence of rheumatic heart disease in Indigenous communities is amongst the highest in the world [4, 15]. Several recent studies provide molecular evidence of the scabies mite itself promoting streptococcal growth in pyoderma through complement inhibitors [1, 16, 17]. Association of scabies with these long-term health problems makes it an important factor to consider in Indigenous health. Despite this, there is a relative paucity of genetic and genomic information on scabies [1].

The scabies mite belongs to the subclass Acari (Arthropoda: Chelicerata: Arachnida: Acari), which contains around 48,000 species of ticks and mites [18], and order Sarcoptiformes. Currently, the scabies mite is classified taxonomically as a single species with different varieties

based on host specificity [19]. Reportedly, cross infectivity is rare and when it happens it is generally of temporary nature and self-limiting infestation [19, 20]. However, evidence regarding host specificity is conflicting, with some studies suggesting cross infectivity is possible between certain animal and human varieties [21, 22], while others suggest mono-specificity of scabies mite varieties [4, 5, 23–26]. For example, genetic and phylogenetic studies that used hypervariable satellite markers, 16S rRNA and cytochrome oxidase subunit I of mitochondria (cox1), have shown that human and dog mites are genetically distinct in north Australian sympatric populations and that gene flow between the scabies mite population is extremely rare [19, 27]; while another study, using the cox1 gene showed that dog mites from China, USA and Australia are genetically similar to with certain human mites from Australia [21, 22]. Choice of the genetic marker may play a role in conflicting outcomes. Additionally, it has been postulated that each variety, when infesting a non-native host, can acquire morphological and innate characteristics suited to the host through selection if it is allowed to persist due to immunodeficiency or malnutrition of the host [20].

Host specificity has been an area of ongoing investigation because potential cross infectivity, in particular between domestic and companion animals and humans, would have important implications for disease control programs. There are no morphological differences between mites from different host species [20], but multiple failed experimental cross infestation attempts indicated that physiological differences may exist [1, 27, 28]. Mites from dogs have successfully established long term infestations in rabbits [28]. Microsatellite studies in wild animal mite populations also indicated a limited gene flow between mites from sympatric host populations [29], but a further study suggested a prey to predator transfer of mites may be possible [30]. Taken together, available data suggests that a limited gene flow occurs between host-associated populations of scabies mites, however a strict 'host taxon' law cannot be assumed.

Here, we describe the *in silico* isolation, *de novo* assembly and analysis of the mitochondrial genome of scabies mites from a variety of complex metagenomic samples. The samples consisted of thousands of whole mites collected from two clinical isolates from different regions of Australia and replicates from a laboratory porcine model of scabies [31]. We assembled the mitochondrial genome by applying a bespoke, iterative bait-and-assemble strategy to the massively parallel sequencing data from each mixture—demonstrating the utility of the general approach on complex metagenomics mixtures. Additionally, we identified 6 mitochondrial haplotype clusters in the human (*Sarcoptes scabiei* var. *hominis*) and pig (*Sarcoptes scabiei* var. *suis*) scabies/mange mite populations sampled. This allowed us to examine the genetic diversity within and between isolates and suggests an extremely low level of diversity overall. To the best of our knowledge, these findings provide the first view of the genetic diversity of individual scabies infestations (intra-host diversity) based on whole mitochondrial genome sequences.

Methods

Collection and DNA extraction

The collection of human patient samples was approved by the Human Research Ethics Committee of the Northern Territory Department of Health and Menzies School of Health Research (approval 13–2027) and informed consent was obtained in writing from each participant. Animal care and handling procedures used in this study followed the Animal Care and Protection Act, in compliance with the Australian code of practice for the care and use of animals for scientific purposes, outlined by the Australian National Health and Medical Research Council. The study was approved by the QASP and the QIMR Berghofer MRI Animal Ethics Committees (DEEDIAEC SA2012/02/381, QIMR A0306-621M).

Skin scrapings were collected from two unrelated patients with severe crusted scabies (patient A and patient B) from different regional areas of Northern Territory, Australia. The

collections from individual patients were made 14 months apart and were likely independent. On both occasions, scabies mites (*S. scabiei* var. *hominis*) were individually picked from the skin. Each sample contained >1000 mites. Two samples of pig mites (*S. scabiei* var. *suis*) were collected from an inbred population of mites from a pig model [<u>31</u>]. The samples were taken from different pigs from consecutive cohorts (where infections are passed on to new piglets from a previous group) at different time points. The first sample consisted of >1000 whole adult mites (pig unwashed). The second sample also consisted of >1000 mites, but was split into three subsamples, which were washed using different protocols to reduce the amount of bacteria present on the surface of the mites due to the wound micro-environment (pig washed 1, 2 and 3 respectively). The protocol entailed: (1) 15 min wash at room temperature in 4% paraformaldehyde in water [<u>32</u>]; (2) 1 hour incubation at 37°C in 150 mM NaCl, 10 mM EDTA, pH8.0, 0.6% SDS, and 0.125 ug/ul lysozyme [adapted from <u>33</u>] and; (3) 1 hour incubation at 37°C in 1% bleach (Sodium hypochlorite) in water. Mites were subsequently rinsed twice in water. Between wash steps mites were centrifuged at 10000 rpm for 2 min.

Whole genome sequencing

Whole mites were crushed and DNA was extracted from each sample using a Blood and Cell culture DNA Kit QIAGEN and a modified procedure, adapted from the manufacturer's protocol. Washed mites were submerged in 1 ml of ice-cold lysis buffer (20 mM EDTA, 100 mM NaCL, 1% TritonX-100, 500 mM Guanidine-HCl, 10 mM Tris pH7.9) and homogenized with stainless steel beads of 2.8mm diameter at 6800rpm, 3 cycles, 30 sec per cycle, 30 sec between cycles. The suspension of lysed mites was supplemented with DNase free RNase A to 0.2 mg/ ml and with Proteinase K to 0.8 mg/ml and incubated at 50°C for 1.5 h. After centrifugation at 4000g for 10 min to pellet insoluble debris the genomic DNA was isolated on the QIAGEN genomic tip as instructed in the manufacturer's protocol. Six DNA libraries were constructed and 100 nucleotide (nt) long paired-end reads were generated on an Illumina HiSeq 2500. Additionally, 54 S. scabiei var. suis eggs were collected from the pig model. To reduce any surface bacteria, these were washed washed twice in 4% paraformaldyde and rinsed twice in water. DNA was extracted separately from 51 individual eggs, and from 3 pools of 5 eggs and 1 pool of 16 eggs. Sequencing libraries were constructed using the Nugen Ovation SP Ultralow DNA kit for 3 single eggs, and pools of 5 and 16 eggs, and sequenced using an Illumina HiSeq 2500 with 2x100 nt reads. Raw data is available via ENA accession PRIEB12428.

Initial de novo assembly

Sequence read quality was assessed using FASTQC [34]. Adapter and quality (Q \geq 20) trimming was performed using TrimGalore! (v0.3.1) [35]. For the human and unwashed pig samples, reads were *de novo* assembled using Velvet (v1.2.08) [36]. To establish the best k-mer size, we initially assembled all reads from patient B using k-mer values of k = 61, 63, 65, 67, 69, 71, 73, 75, 79, 85, 89 and 95. Based on the quality of these assemblies, we used k = 69, 75, 77, 79, 81, 83, 85, 89 and 95 for patient A and the unwashed pig samples.

Metagenomic profiling

Metagenomic profiling of the patient B mite assembly was carried out using PhymmBL (v4.0) [<u>37</u>]. We augmented the bundled PhymmBL model database with interpolated Markov models trained on the spider mite, *Tetranychus urticae* (strain London) (available at <u>https://bioinformatics.psb.ugent.be/gdb/tetranychus/</u>) [<u>38</u>] as a proxy for scabies mite. To estimate the species abundance, reads were aligned back to contigs using Bowtie2 (v2.2.3) in local alignment mode [<u>39</u>] and the number of reads aligning to each contig were counted.

The Mitochondrial Genome Sequence of the Scabies Mite



Fig 1. Bait-and-assembly workflow for metagenomics data. Two iterations were performed. NT stands for nucleotide collection blast database (nt) from NCBI.

doi:10.1371/journal.pntd.0004384.g001

De novo assembly of the mitochondrial genome

For the two human samples and the unwashed pig sample, the mitochondrial genomes were assembled individually using a bait-and-reassemble strategy (see Fig 1 for overview). First, contigs from the whole genome assemblies were aligned to European house dust mite (EHDM), Dermatophagoides pteronyssinus mitochondrial reference genome (NCBI GenBank: EU884425) [40] using LASTZ (v1.02.00) [41] with default settings. For patient B, contigs from the k = 65 assembly were used for alignment; for patient A, contigs from k = 77, 79 and 89 were used; and for the unwashed pig sample, contigs from the k = 69 and 81 assemblies were used. These different k-mer value assemblies varied in their N50 values, largest contigs sizes and median read coverage. The aligned contigs were then filtered against the National Center for Biotechnology Information (NCBI) NT database for scabies mite mitochondrial contigs using BLASTN (v2.2.29) [42] to remove likely false positives due to homology with other mitochondrial genomes (e.g. host mitochondrial genome). After the filtering step, contigs from patient B k = 65 assembly gave the best coverage of EHDM mitochondrial reference genome. Adapter and quality trimmed reads from the patient B sample were then aligned back to the filtered patient B k = 65 contigs using Bowtie2 (v2.2.3) in local alignment mode. Aligned reads were then used to de novo assemble the patient B scabies mite mitochondrial genome using Velvet

(v1.2.08). Assemblies were run using k = 69, 79, 87, 89, 91 and 95. A k-mer size of 91 gave the longest single contig and was selected as the reference mitochondrial genome of scabies mite.

To generate the mitochondrial reference for patient A, adapter and quality trimmed reads from the patient A sample were aligned to the patient B mitochondrial reference genome, and matching reads were *de novo* assembled using Velvet. The final contigs were scaffolded and gaps filled with Ns. This scaffold was then used to realign the patient A reads and reassemble. This process was run iteratively. The second iteration gave the minimal number of Ns in the contig and this was retained as the patient A mite mitochondrial genome.

The same process was used for the unwashed pig sample, but in this case the first iteration gave the best contig with minimal Ns.

Annotation

Mitochondrial protein-coding genes were predicted on patient B mitochondrial reference genome using the MITOS pipeline (accessed online, August 2014) [43] using the invertebrate mitochondrial genetic code (with default settings). To verify the annotations, open reading frames (ORFs) between stop to stop codons were extracted from the assembled patient B mite genome using the GETORF program from the EMBOSS package (v6.6.0) [44] using the invertebrate mitochondrial codon table. The extracted 774 ORFs were then used to search the NCBI NR protein database using BLASTP (v2.2.29) (E-value threshold = 0.1). For further verification, 13 protein-coding genes from EHDM were aligned to the patient B mite reference genome using TBLASTN (v2.2.29).

12S and 16S ribosomal RNA were annotated using covariance models for those genes using Infernal (v1.1rc4) [45, 46]. The covariance models were built from multiple sequence alignments of 12S and 16S rRNA genes from three sarcoptiformes, *Steganacarus magnus* (NCBI RefSeq: NC_011574), *Dermatophagoides pteronyssinus* (NCBI GenBank: EU884425) and *Dermatophagoides farinae* (NCBI RefSeq: NC_013184) using Clustal-omega (run on the web version on 15/12/2013) [47]. MITFI (within the MITOS pipeline) was used to identify 21 of 22 standard mitochondrial tRNA genes.

Single nucleotide polymorphisms and diversity

For each of the six samples, reads from each sample were aligned to the patient B reference genome using the Bowtie2 aligner in local mode. Pileups were generated using SAMtools (v0.1.19-44428cd) [48] mpileup. Varscan (v2.3.6) [49] was then used to call SNPs with the mpileup2snp command (Min Coverage: 300; Min Variant Frequency: 0.01; otherwise default parameters). SNPs called within the 100 nt boundary at both ends of the genome (positions 1–100 and 13,820–13,919) were manually filtered out due to lack of reliability of alignments at those low complexity regions. For each sample, SNP frequency was estimated using the ratio of reads supporting the variant to total read depth at the SNP.

Results

Sequencing of the two human and four pig samples (unwashed and 3 washed technical replicates) generated 46 (patient B) to 62 (pig washed 1) million read pairs per sample (<u>S1 Table</u>). As expected, the libraries also contained host and microbial DNA. We estimated the level of contamination using two approaches. PhymmBL was applied to the patient B mite mitochondrial assembly to classify contigs into taxa. This revealed that 44% of contigs were from bacteria or other non-arachnid species (<u>S1A Fig</u>). Species abundance was estimated by re-aligning the reads back to the contigs. This revealed that 6% of the reads were derived from contaminants (<u>S1B Fig</u>). Using our bait-and-assemble strategy, we assembled the mitochondrial genome of each sample (ENA accessions: LN874268-LN874270). The patient B mitochondrial genome assembly comprised a single contig of length 13,919 nt. The patient A and unwashed pig assemblies were 13,902 and 14,044 nt respectively. Both also consisted of single contigs, but each contained a contiguous gap of 118 (patient A) to 255 (unwashed pig) Ns. We were unable to generate a circular mitochondrial genome sequence and found the contig ends were composed of repeat rich sequences. Assuming the scabies mitochondrial genome is circular, we estimate that an AT-rich gap of approximately 300 nt in length is present in the human mite reference assembly. Similar gaps are also present in the other assemblies.

Similar to other sequenced Acari mitochondrial genomes, the scabies mite mitochondrial genome is also highly AT-rich with a 19.32% GC-content. The strand bias is characterized by GC and AT skews, calculated using (G%-C%)/(G%+C%) and (A%-T%)/(A%+T%) respectively. The GC skew for the leading strand is -0.0327 and AT skew is 0.0341. Negative GC skew and positive AT skew are similar to standard metazoan genome strand biases including Acari. Reversal in GC skew in terms of strands has been observed in only two mites so far, *Varroa destructor* [40, 50] and house dust mite [40].

Annotation

Thirty-five predicted genes were identified in the scabies mite (var. *hominis*) mitochondrial genome (Fig 2). These include 13 protein coding genes, two ribosomal RNA genes, and 20 tRNA genes. All of the expected mitochondrial genes for a typical metazoan mitochondrial genome were identified, except for two tRNAs (Alanine and Tyrosine). Search with BLASTP of extracted stop-to-stop ORFs from the mitochondrial genome also verified 12 of the 13 protein-coding genes (except the ATP8 gene) and TBLASTN alignment of 13 protein coding genes from the EHDM mitochondrial genome to the scabies mite mitochondrial genome also confirmed 12 of the 13 protein coding genes (except the ND4L gene) in the scabies mite mitochondrial genome. Two tRNA genes identified by tRNAscan-SE were considered ambiguous due to their overlap with other genes (A is overlapping with C and Y is overlapping with NAD4L). However, the overlap is not on the same strand.

The gene order and strand specification of the protein coding and ribosomal RNA genes are the same as that of EHDM. All identified tRNA genes, except tRNA-C and tRNA-V also maintain the same gene order and strand as EHDM. The tRNA genes that are not syntenic with EHDM (tRNA-C, tRNA-V) are also on the opposite strand to that of EHDM.

Single nucleotide polymorphisms and mitochondrial haplotypes

To identify genetic polymorphisms in the sequenced mite pools, reads were mapped back to the reference Mt genome (patient B). The average depth of coverage was 2914 across samples (average per sample ranging 1698–4299). A total of 665 single nucleotide polymorphisms (SNPs) were identified across all samples relative to the reference Mt genome (patient B assembly): 601 SNPs in the patient A sample, 598 SNPs in the patient B sample, and 102 SNPs in the unwashed pig sample, while the washed pig samples (w1, w2, w3) contained 102, 100, 102 SNPs respectively (<u>S2 Table</u>). The four pig mite samples are effectively replicates (biological and technical) and were highly concordant.

Within each sample, SNP allele frequencies, estimated from the ratio of reads supporting the variant to the total coverage, were tightly grouped into a small number of clusters (Fig 3A), suggestive of the presence of just a few mitochondrial haplotypes in each sample. To estimate the number and frequencies of haplotypes, and to infer their sequences, we used k-means clustering (S3 Table). SNPs common to all haplotypes within a sample have a frequency of 1.

PLOS | NEGLECTED TROPICAL DISEASES



gene tRNA rRNA

Length: 13,919 bp

GC Content: 19.32%

Fig 2. Annotation of the scabies mite, Sarcoptes scabiei (var. hominis), mitochondrial genome. tRNA genes are abbreviated using the one letter amino acid code. The inner brown plot is plotted for GC content using window size of 500 and step 1. The plot shows deviations from the average GC content of the genome (0.1932). The plot is scaled based on the maximum (0.3034) and minimum (0.1078) GC content values.

doi:10.1371/journal.pntd.0004384.g002

Additional haplotypes are defined by the presence of extra SNPs with mean frequency less than 1. We also examined reads supporting SNPs located close to each other (<100 nt) for evidence that SNPs were on the same haplotype (supported by the same reads) or distinct haplotypes (never co-occurring on one read). Only 4 pairs of SNP were located within 100 nucleotides of each other and in two different frequency clusters. In each sample, we gave the haplotype with the highest frequency the prefix H1. Haplotypes with successively lower frequencies are labeled H2 and H3, in that order.



Fig 3. Scabies haplotype groups. (A) SNP allele frequencies in each sample show strong clustering and define haplotypes. (B) Phylogenetic tree of the inferred haplotypes from multiple human and pig samples group into 6 haplogroups and 2 broad clades (red and blue boxes).

doi:10.1371/journal.pntd.0004384.g003

In the patient B sample, a single clear cluster in the SNP frequencies was observed (Fig_3A). This implies the presence of two haplotypes. The dominant haplotype, H1_B_REF, has an estimated frequency of 0.98 and corresponds to the consensus obtained from *de novo* assembly. The second haplotype (H2_B) is defined by the presence of 598 SNPs and has an estimated frequency of 0.02.

The patient A sample contains a single clear cluster, containing 593 SNPs with a frequency of 1.0. An additional 8 SNPs do not cluster well with the dominant group. These fall into four other clusters identified by k-means. We chose to ignore the three smallest clusters, resulting in just two closely-related haplotypes: H1_A, which has a frequency of 0.98, and H2_A, which has a frequency of 0.02 and is separated from H1_A by just 4 SNPs. The lack of replication of these SNPs makes the significance of these closely related haplotypes unclear.

All four pig samples contain three clear clusters, forming three haplotypes in each sample. The haplotypes with the highest frequencies (H1_u, H1_w1, H1_w2, H1_w3) have estimated frequencies of 0.89, 0.91, 0.89 and 0.90 respectively. Additional haplotypes are composed of combinations of the clusters (S3 Table) and are concordant between samples. The second haplotypes have estimated frequencies of 0.10, 0.07, 0.09 and 0.09 for the unwashed and three washed samples respectively and we label these H2_u, H2_w1, H2_w2, and H2_w3. The third haplotypes, which we label H3_u, H3_w1, H3_w2 and H3_w3 for the unwashed and three washed samples respectively, have estimated frequencies 0.02, 0.01, 0.02 and 0.01 respectively. The H1_u, H1_w1, H1_w2, H1_w3 haplotypes have 82, 82, 81, 81 SNPs; H2_u, H2_w1, H2_w2, H2_w3 have 82, 82, 81, 81 SNPs; and H3_u, H3_w1, H3_w2, H3_w3 have 87, 87, 85, 87 SNPs respectively.

Similar analyses of DNA sequencing data from a limited number of individual scabies mite eggs or small pools did not identify clusters in the SNP allele frequencies (<u>S2 Fig</u>), suggesting that multiple haplotypes provide evidence for genetic diversity rather than heteroplasmy.

To understand the relationship between the 16 inferred haplotypes, we constructed a phylogenetic tree based on the SNPs present in each haplotype sequence using MEGA5 (v5.2.2) [51] with a distance measure based on the number of differences between haplotype sequences (S4 <u>Table</u>). The tree shows that the haplotypes fall into two broad clades and six haplogroups (Fig <u>3B</u>). Three haplogroups are found in clinical isolates (human mite haplogroups 1–3), while the pig mites comprise three haplogroups (pig mite haplogroups 1–3). The average difference between haplotypes within each of the pig mite haplogroups is 1–2 SNPs. The two haplotypes in human mite haplogroup 3 are almost identical and very similar to human mite haplogroup 2, forming one of the clades, while human mite haplogroup 1 is distinct from the other human mite haplogroups, and more similar to the pig mite haplogroups, forming the second clade.

Discussion

We *de novo* assembled the mitochondrial genome of the scabies mite using massively parallel sequencing data from thousands of pooled whole mites obtained from two clinical isolates from different parts of northern Australia and from a laboratory pig model. Our approach was to initially perform a full metagenomic *de novo* assembly. As expected, the samples were contaminated by bacterial reads presumably from the scabies mite gut and the scabies lesion micro-environment. We then iteratively selected Mt contigs and used these as bait to recruit and assemble genuine mite reads. Our bespoke *de novo* assembly approach has some similarity to an existing bait-and-assembly method called MITObim [52]. MITObim first directly baits the short reads using a closely related mitochondrial genome, then iteratively maps reads and performs contig extension using MIRA. De novo assembly is also supported, but only for "well behaved" data. In contrast, our approach performs full de novo assembly of the metagenomic mixture, baiting of contigs using the house dust mite genome and filtering of host

mitochondrial contigs, then 1–2 iterations of read alignment and de novo assembly using velvet. Our results substantially extend the limited case studies and simulations used to validate MITObim and demonstrate that the general approach of mitochondrial genome bait-andassembly works on real examples that are highly complex metagenomic mixtures involving both genetic heterogeneity and host/bacterial contamination.

Mitochondrial genomes are generally circular, but rare occurrences of linear mitochondrial genomes have been reported [53]. While we were unable to circularise the scabies mite mitochondrial genome assemblies, it seems unlikely that it is linear; a simpler explanation is that the region contains highly repetitive AT-rich sequences (which we observe on the flanking regions of the break) and these are difficult to map reads to and assemble across. The same region in EHDM mitochondrial genome contains the origin of replication site. This might also be a factor for the missing coverage in this region.

The gene content and organisation of the *Sarcoptes scabiei* var. *hominis* mitochondrial genome is the same as the recently published var. *canis* mitochondrial genome contig (NCBI GenBank: CM003133 JXLN01000000) [54]. The gene order is also similar to the rabbit ear mite, *Psoroptes cuniculi*, mitochondrial genome (NCBI Refseq: NC_024675) [55] with the exception of the two additional tRNA genes found in the latter (Fig 4) and the house dust mite (NCBI GenBank: EU884425), suggesting a close evolutionary relationship between these species. These observations further support the hypothesized close evolutionary relationship between the parasitic scabies mite and free-living house dust mite [56, 57]. However, it is distinct from most other Acari mitochondrial genomes sequenced to date [40], including *Tetranychus urticae* (NCBI GenBank: EU345430). It is also quite distinct from *Limulus polyphemus* (NCBI GenBank: NC_003057), which represents the arthropod ground pattern of gene arrangement for chelicerates [58].

We found variations in the diversity of infestations between individuals (clonal or heterogeneous). Mites from skin scrapings from one patient with crusted scabies (patient A) appeared to be essentially clonal. This lack of diversity suggests that either a single female mite may initiate infection, or selection by the host immune system may operate. Mites from the second patient (patient B), who had more severe crusted scabies than patient A, showed two highly divergent haplotypes. Remarkably, one of the haplotypes in the patient B sample was much more similar to the pig mite haplotypes than to the other human mite haplotypes. This may reflect an actual cross-species infection, highlight the possibility of this, or could merely suggest that the genetic differences between varieties are limited.

The population-level genetic diversity of scabies mites has been studied extensively using single or small numbers of gene sequences. One recent study based on cox1 gene sequences showed that scabies infestations of humans, dogs and other animals fall into 3 clades [21]. Two clades contained only geographically isolated var. *hominis* mites, while the third contained closely related human and animal mites. Another recent study, which used 3 gene sequences, reported 5 clades—4 distinct var. *hominis* clades (one for each geographical region studied) and one clade containing closely-related var. *hominis* and other animal scabies mite varieties [21, 22]. These findings are consistent with our observation that one of the patient B mite haplotypes was more similar to the pig mite haplotypes than to the other human mite haplotypes. Since we used the entire mitochondrial genome, we expect our results to be robust. Moreover, our intra-sample inference of haplotypes has revealed that patients and animals can harbour multiple genetically divergent mites, or near clonal infestations. To the best of our knowledge, this is the first view of intra-host genetic diversity for scabies mites.

One issue with our analysis of haplotypes is that we are unable to utilize a few unclustered and unreplicated SNPs, in particular, from the patient B sample. These may define additional haplo-types or represent genetic drift. Notwithstanding, these haplotypes would be extremely closely

PLOS | NEGLECTED TROPICAL DISEASES





are drawn according to order, intergenic distances are not shown and genes sizes are not drawn according to scale. J stands for majority strand and N stands for minority strand. *L. polyphemus* is included in the comparison representing the arthropod ground pattern of gene arrangement for chelicerates. Yellow boxes represent genes that have different positions relative to scabies mite; teal boxes represent genes that are different in terms of both position and orientation; and orange boxes represent genes that are not present in scabies mite. Accession numbers of the Mt genomes are listed in <u>S5 Table</u>.

doi:10.1371/journal.pntd.0004384.g004

related. Our justification is that these are very small differences between haplotypes and may be due to sequencing errors and thus erroneously detected variants, whereas large numbers of SNPs with clustered frequency are more likely to be real. Regardless, our analysis provides a first look at the genetic diversity of scabies infections within patients. However, further study with broader population sample is required to refine the intra- and inter-host population diversity structure.

Conclusion

Scabies is responsible for significant morbidity in Indigenous Australians in many remote communities of northern and central Australia. Effective control and prevention of scabies epidemics in those communities is of paramount importance as scabies has a long-term effect on the life expectancy and quality. We have sequenced, assembled and annotated the mitochondrial genome of the scabies mite using a bespoke bait-and-assembly approach; we identified SNPs in multiple isolates from patients and a laboratory pig model, and inferred the haplotype structure and diversity of individual infections. We used these tools to investigate the genetic diversity within individual infestations. Larger samples are now needed.

The development of genomics resources for studying the scabies mite will accelerate research into this parasite, just as genome sequences have for other neglected parasitic diseases.

For example, in malaria genomic resources provided means to identify drug resistance causing mutations [59] or in schistosomiasis it helped suggesting new approaches to preventions and strategies for control [60]. The scabies mite mitochondrial genome sequence will facilitate further population genetics research in this area.

Supporting Information

S1 Fig. Metagenomic profile of the patient B sample. Plots generated using PhymmBL [<u>37</u>] and Krona [<u>61</u>]. (A) The patient B assembly has nearly half of the contigs belonging to contaminants. (B) Realigning reads back to contigs shows species abundance in the patient B sample reads used in the assembly.

(TIF)

S2 Fig. SNP allele frequencies in sequence data from single and small pools of scabies mite eggs (*S. scabiei* var. *suis*). Short read sequencing data of DNA from single eggs, or pools of 5 or 16 eggs were aligned to the scabies mite Mt reference genome (patient B), and SNPs called using the method described previously (except the minimum coverage threshold was set to 15 in this case due to lower coverage). The low frequency SNPs in the pooled egg samples (5 and 16) are all adjacent to homopolymer runs and consistent to be sequencing errors. The cluster of SNPs with frequency 1.0 and absence of other clear clusters suggests that only a single haplo-type is present in individual eggs and that genetic diversity, rather than heteroplasmy explains the clusters in other samples.

(TIF)

S1 Table. Details of sequencing libraries. (XLSX)

S2 Table. Single nucleotide polymorphisms called by Varscan on all samples along with their corresponding base positions. (TXT)

S3 Table. Results from k-means clustering showing SNP frequencies and formation of haplotypes based on frequency clustering. (XLSX)

S4 Table. Pairwise distance between haplotypes in a matrix format. Here, distance is represented by number of base differences between haplotype sequences. (XLS)

S5 Table. Accession numbers of Mt genomes used for gene arrangement comparison. (XLSX)

Acknowledgments

We acknowledge Mr. Andrew Kelly and Mrs. Beverly Hutchinson for the animal management at the Queensland Animal Science Precinct (QASP).

Author Contributions

Conceived and designed the experiments: EM DH BJC KF ATP. Performed the experiments: DH KF. Analyzed the data: EM ATP. Contributed reagents/materials/analysis tools: EM TS MB DH BJC KF ATP. Wrote the paper: EM TS MB DH BJC KF ATP.

References

- Walton SF, Holt DC, Currie BJ, Kemp DJ. Scabies: New Future for a Neglected Disease. 2004; 57:309–76. doi: <u>10.1016/s0065-308x(04)57005-7</u> PMID: <u>15504541</u>
- Holt DC, McCarthy JS, Carapetis JR. Parasitic diseases of remote Indigenous communities in Australia. International journal for parasitology. 2010; 40(10):1119–26. doi: <u>10.1016/j.ijpara.2010.04.002</u> PMID: <u>20412810</u>.
- Roberts LJ, Huffam SE, Walton SF, Currie BJ. Crusted scabies: clinical and immunological findings in seventy-eight patients and a review of the literature. The Journal of infection. 2005; 50(5):375–81. doi: 10.1016/j.jinf.2004.08.033 PMID: 15907543.
- Currie BJ, Carapetis JR. Skin infections and infestations in Aboriginal communities in northern Australia. Australasian journal of dermatology. 2000; 41(3):139–43. PMID: <u>10954983</u>
- Holt DC, Fischer K. Novel insights into an old disease: recent developments in scabies mite biology. Current opinion in infectious diseases. 2013; 26(2):110–5. doi: <u>10.1097/QCO.0b013e32835eb986</u> PMID: <u>23385639</u>.
- Bornstein S, Mörner T, Samuel WM. Sarcoptes scabiei and sarcoptic mange. Parasitic diseases of wild mammals. 2001:107–19.
- Samuel W, Pybus M, Kocan A. Parasitic Diseases of Wild Mammals Iowa State University Press. Ames, IA. 2001.
- Clucas DB, Carville KS, Connors C, Currie BJ, Carapetis JR, Andrews RM. Disease burden and health-care clinic attendances for young children in remote aboriginal communities of northern Australia. Bull World Health Organ. 2008; 86(4):275–81. Epub 2008/04/29. S0042-96862008000400012 [pii]. PMID: <u>18438516</u>; PubMed Central PMCID: PMC2647416.
- Romani L, Steer AC, Whitfeld MJ, Kaldor JM. Prevalence of scabies and impetigo worldwide: a systematic review. The Lancet Infectious Diseases. 2015.
- Allen AM, Taplin D. Skin infections in eastern Panama. Survey of two representative communities. The American journal of tropical medicine and hygiene. 1974; 23(5):950–6. PMID: <u>4451236</u>
- Kelly C, Taplin D, Allen AM. Streptococcal Ecthyma: Treatment With Benzathine Pencillin G. Archives of dermatology. 1971; 103(3):306–10. PMID: <u>5548277</u>
- Bowen AC, Tong SY, Andrews RM, O'Meara IM, McDonald MI, Chatfield MD, et al. Short-course oral co-trimoxazole versus intramuscular benzathine benzylpenicillin for impetigo in a highly endemic region: an open-label, randomised, controlled, non-inferiority trial. The Lancet. 2014; 384(9960):2132– 40.
- Carapetis JR, McDonald M, Wilson NJ. Acute rheumatic fever. The Lancet. 2005; 366(9480):155–68. doi: 10.1016/s0140-6736(05)66874-2
- Marshall CS, Cheng AC, Markey PG, Towers RJ, Richardson LJ, Fagan PK, et al. Acute post-streptococcal glomerulonephritis in the Northern Territory of Australia: a review of 16 years data and comparison with the literature. The American journal of tropical medicine and hygiene. 2011; 85(4):703–10. doi: 10.4269/ajtmh.2011.11-0185 PMID: 21976576
- Carapetis JR, Wolff DR, Currie BJ. Acute rheumatic fever and rheumatic heart disease in the top end of Australia's Northern Territory. The Medical journal of Australia. 1996; 164(3):146–9. PMID: 8628132
- Mika A, Reynolds SL, Pickering D, McMillan D, Sriprakash KS, Kemp DJ, et al. Complement inhibitors from scabies mites promote streptococcal growth–a novel mechanism in infected epidermis? PLoS neglected tropical diseases. 2012; 6(7):e1563. doi: <u>10.1371/journal.pntd.0001563</u> PMID: <u>22815998</u>
- Mika A, Goh P, Holt DC, Kemp DJ, Fischer K. Scabies mite peritrophins are potential targets of human host innate immunity. PLoS Negl Trop Dis. 2011; 5(9):e1331. doi: <u>10.1371/journal.pntd.0001331</u> PMID: <u>21980545</u>; PubMed Central PMCID: PMC3181238.
- Harvey MS. The neglected cousins: what do we know about the smaller arachnid orders? Journal of Arachnology. 2002; 30(2):357–72.
- Walton S, Dougall A, Pizzutto S, Holt D, Taplin D, Arlian L, et al. Genetic epidemiology of Sarcoptes scabiei (Acari: Sarcoptidae) in northern Australia. International journal for parasitology. 2004; 34 (7):839–49. PMID: 15157767
- Fain A. Epidemiological problems of scabies. International Journal of Dermatology. 1978; 17(1):20–30. PMID: <u>415012</u>
- Andriantsoanirina V, Ariey F, Izri A, Bernigaud C, Fang F, Charrel R, et al. Sarcoptes scabiei mites in humans are distributed into three genetically distinct clades. Clinical Microbiology and Infection. 2015.
- Zhao Y, Cao Z, Cheng J, Hu L, Ma J, Yang Y, et al. Population identification of Sarcoptes hominis and Sarcoptes canis in China using DNA sequences. Parasitology research. 2015; 114(3):1001–10. doi: 10.1007/s00436-014-4266-1 PMID: 25547078

- Alasaad S, Soglia D, Spalenza V, Maione S, Soriguer RC, Pérez JM, et al. Is ITS-2 rDNA suitable marker for genetic characterization of Sarcoptes mites from different wild animals in different geographic areas? Veterinary parasitology. 2009; 159(2):181–5. doi: <u>10.1016/j.vetpar.2008.10.001</u> PMID: <u>19019542</u>
- Berrilli F D'Amelio S, Rossi L. Ribosomal and mitochondrial DNA sequence variation in Sarcoptes mites from different hosts and geographical regions. Parasitology research. 2002; 88(8):772–7.
- Gu X-B, Yang G-Y. A study on the genetic relationship of mites in the genus Sarcoptes (Acari: Sarcoptidae) in China. International Journal of Acarology. 2008; 34(2):183–90.
- Zahler M, Essig A, Gothe R, Rinder H. Molecular analyses suggest monospecificity of the genus Sarcoptes (Acari: Sarcoptidae). International journal for parasitology. 1999; 29(5):759–66. PMID: 10404272
- Walton SF, Choy JL, Bonson A, Valle A, McBroom J, Taplin D, et al. Genetically distinct dog-derived and human-derived Sarcoptes scabiei in scabies-endemic communities in northern Australia. The American journal of tropical medicine and hygiene. 1999; 61(4):542–7. PMID: 10548286
- Arlian LG, Runyan RA, Estes SA. Cross infestivity of Sarcoptes scabiei. J Am Acad Dermatol. 1984; 10 (6):979–86. PMID: 6736342.
- Alasaad S, Oleaga A, Casais R, Rossi L, Min AM, Soriguer RC, et al. Temporal stability in the genetic structure of Sarcoptes scabiei under the host-taxon law: empirical evidences from wildlife-derived Sarcoptes mite in Asturias, Spain. Parasit Vectors. 2011; 4:151. doi: <u>10.1186/1756-3305-4-151</u> PMID: <u>21794141</u>; PubMed Central PMCID: PMC3160406.
- 30. Gakuya F, Rossi L, Ombui J, Maingi N, Muchemi G, Ogara W, et al. The curse of the prey: Sarcoptes mite molecular analysis reveals potential prey-to-predator parasitic infestation in wild animals from Masai Mara, Kenya. Parasit Vectors. 2011; 4:193. doi: <u>10.1186/1756-3305-4-193</u> PMID: <u>21978557</u>; PubMed Central PMCID: PMC3193033.
- Mounsey K, Ho MF, Kelly A, Willis C, Pasay C, Kemp DJ, et al. A tractable experimental model for study of human and animal scabies. PLoS Negl Trop Dis. 2010; 4(7):e756. doi: <u>10.1371/journal.pntd.</u> <u>0000756</u> PMID: <u>20668508</u>; PubMed Central PMCID: PMC2907415.
- Valiente Moro C, Chauve C, Zenner L. Experimental infection of Salmonella Enteritidis by the poultry red mite, Dermanyssus gallinae. Vet Parasitol. 2007; 146(3–4):329–36. doi: <u>10.1016/j.vetpar.2007.02.</u> 024 PMID: 17382475.
- Reed DL, Hafner MS. Phylogenetic analysis of bacterial communities associated with ectoparasitic chewing lice of pocket gophers: a culture-independent approach. Microb Ecol. 2002; 44(1):78–93. doi: 10.1007/s00248-002-0009-4 PMID: 12019462.
- 34. FASTQC. Available from: http://www.bioinformatics.babraham.ac.uk/projects/fastqc/.
- 35. Trim Galore! Available from: http://www.bioinformatics.babraham.ac.uk/projects/trim_galore/.
- Zerbino DR, Birney E. Velvet: algorithms for de novo short read assembly using de Bruijn graphs. Genome research. 2008; 18(5):821–9. doi: <u>10.1101/gr.074492.107</u> PMID: <u>18349386</u>; PubMed Central PMCID: PMC2336801.
- Brady A, Salzberg SL. Phymm and PhymmBL: metagenomic phylogenetic classification with interpolated Markov models. Nature methods. 2009; 6(9):673–6. doi: <u>10.1038/nmeth.1358</u> PMID: <u>19648916</u>; PubMed Central PMCID: PMC2762791.
- Grbić M, Van Leeuwen T, Clark RM, Rombauts S, Rouzé P, Grbić V, et al. The genome of Tetranychus urticae reveals herbivorous pest adaptations. Nature. 2011; 479(7374):487–92. doi: <u>10.1038/</u> nature10640 PMID: 22113690
- Langmead B, Salzberg SL. Fast gapped-read alignment with Bowtie 2. Nature methods. 2012; 9 (4):357–9. doi: 10.1038/nmeth.1923 PMID: 22388286; PubMed Central PMCID: PMC3322381.
- Dermauw W, Van Leeuwen T, Vanholme B, Tirry L. The complete mitochondrial genome of the house dust mite *Dermatophagoides pteronyssinus* (Trouessart): a novel gene arrangement among arthropods. BMC genomics. 2009; 10:107. Epub 2009/03/17. 1471-2164-10-107 [pii] doi: <u>10.1186/1471-2164-10-107</u> PMID: <u>19284646</u>; PubMed Central PMCID: PMC2680895.
- 41. Harris RS. Improved pairwise alignment of genomic DNA. PhD thesis. 2007.
- Altschul SF, Gish W, Miller W, Myers EW, Lipman DJ. Basic local alignment search tool. Journal of molecular biology. 1990; 215(3):403–10. PMID: <u>2231712</u>
- Bernt M, Donath A, Juhling F, Externbrink F, Florentz C, Fritzsch G, et al. MITOS: improved de novo metazoan mitochondrial genome annotation. Molecular phylogenetics and evolution. 2013; 69(2):313– 9. doi: 10.1016/j.ympev.2012.08.023 PMID: 22982435.
- Rice P, Longden I, Bleasby A. EMBOSS: the European molecular biology open software suite. Trends in genetics. 2000; 16(6):276–7. PMID: <u>10827456</u>

- Nawrocki EP, Eddy SR. Infernal 1.1: 100-fold faster RNA homology searches. Bioinformatics. 2013; 29 (22):2933–5. doi: <u>10.1093/bioinformatics/btt509</u> PMID: <u>24008419</u>; PubMed Central PMCID: PMC3810854.
- Nawrocki EP, Kolbe DL, Eddy SR. Infernal 1.0: inference of RNA alignments. Bioinformatics. 2009; 25 (10):1335–7. doi: <u>10.1093/bioinformatics/btp157</u> PMID: <u>19307242</u>; PubMed Central PMCID: PMC2732312.
- Sievers F, Wilm A, Dineen D, Gibson TJ, Karplus K, Li W, et al. Fast, scalable generation of high-quality protein multiple sequence alignments using Clustal Omega. Molecular systems biology. 2011; 7:539. doi: 10.1038/msb.2011.75 PMID: 21988835; PubMed Central PMCID: PMC3261699.
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The Sequence Alignment/Map format and SAMtools. Bioinformatics. 2009; 25(16):2078–9. doi: <u>10.1093/bioinformatics/btp352</u> PMID: <u>19505943</u>; PubMed Central PMCID: PMC2723002.
- Koboldt DC, Zhang Q, Larson DE, Shen D, McLellan MD, Lin L, et al. VarScan 2: somatic mutation and copy number alteration discovery in cancer by exome sequencing. Genome research. 2012; 22 (3):568–76. doi: <u>10.1101/gr.129684.111</u> PMID: <u>22300766</u>; PubMed Central PMCID: PMC3290792.
- Navajas M, Le Conte Y, Solignac M, Cros-Arteil S, Cornuet J- M. The complete sequence of the mitochondrial genome of the honeybee ectoparasite mite Varroa destructor (Acari: Mesostigmata). Molecular biology and evolution. 2002; 19(12):2313–7. PMID: <u>12446822</u>
- Tamura K, Peterson D, Peterson N, Stecher G, Nei M, Kumar S. MEGA5: molecular evolutionary genetics analysis using maximum likelihood, evolutionary distance, and maximum parsimony methods. Molecular biology and evolution. 2011; 28(10):2731–9. doi: <u>10.1093/molbev/msr121</u> PMID: <u>21546353</u>; PubMed Central PMCID: PMC3203626.
- Hahn C, Bachmann L, Chevreux B. Reconstructing mitochondrial genomes directly from genomic nextgeneration sequencing reads—a baiting and iterative mapping approach. Nucleic acids research. 2013; 41(13):e129. doi: <u>10.1093/nar/gkt371</u> PMID: <u>23661685</u>; PubMed Central PMCID: PMC3711436.
- Nosek J, Tomáška LUr, Fukuhara H, Suyama Y, Kováč L. Linear mitochondrial genomes: 30 years down the line. Trends in Genetics. 1998; 14(5):184–8. PMID: <u>9613202</u>
- Rider SD Jr, Morgan MS, Arlian LG. Draft genome of the scables mite. Parasites & vectors. 2015; 8 (1):1–14.
- 55. Gu X-B, Liu G-H, Song H-Q, Liu T-Y, Yang G-Y, Zhu X-Q. The complete mitochondrial genome of the scab mite Psoroptes cuniculi (Arthropoda: Arachnida) provides insights into Acari phylogeny. Parasites & vectors. 2014; 7(1):1–10.
- 56. Klimov PB, Oconnor BM. Improved tRNA prediction in the American house dust mite reveals widespread occurrence of extremely short minimal tRNAs in acariform mites. BMC genomics. 2009; 10:598. doi: 10.1186/1471-2164-10-598 PMID: 20003349; PubMed Central PMCID: PMC2797822.
- Klimov PB, OConnor B. Is permanent parasitism reversible?—Critical evidence from early evolution of house dust mites. Systematic biology. 2013:syt008.
- Staton JL, Daehler LL, Brown WM. Mitochondrial gene arrangement of the horseshoe crab Limulus polyphemus L.: conservation of major features among arthropod classes. Molecular biology and evolution. 1997; 14(8):867–74. PMID: <u>9254925</u>
- Cheeseman IH, Miller BA, Nair S, Nkhoma S, Tan A, Tan JC, et al. A major genome region underlying artemisinin resistance in malaria. Science. 2012; 336(6077):79–82. doi: <u>10.1126/science.1215966</u> PMID: <u>22491853</u>
- Young ND, Jex AR, Li B, Liu S, Yang L, Xiong Z, et al. Whole-genome sequence of Schistosoma haematobium. Nature genetics. 2012; 44(2):221–5. doi: <u>10.1038/ng.1065</u> PMID: <u>22246508</u>.
- Ondov BD, Bergman NH, Phillippy AM. Interactive metagenomic visualization in a Web browser. BMC bioinformatics. 2011; 12(1):385.